

Article



Algorithm for Application of a Basic Model for the Data Envelopment Analysis Method in Technical Systems

Mariia Pokushko ^{1,2,3,*}, Alena Stupina ^{1,3,4}, Inmaculada Medina-Bulo ², Svetlana Ezhemanskaya ¹, Roman Kuzmich ¹, and Roman Pokushko ³

- ¹ Institute of Business Process Management, Siberian Federal University, Krasnoyarsk 660041, Russia
- ² Superior Engineering School, University of Cadiz, Puerto Real, 11519 Cadiz, Spain
- ³ Institute of Informatics and Telecommunications, Reshetnev Siberian State University of Science and Technology, Krasnoyarsk 660037, Russia
- ⁴ Siberian Fire and Rescue Academy of GPS MES, Zheleznogorsk, Severnaya 662972, Russia
- Correspondence: mari.pokushko@alum.uca.es

Abstract: The aim of this study is to solve the problem of increasing the efficiency of fuel and energy complex enterprises. Because such enterprises are complex systems, it is difficult to optimize their work, taking into account all the technical indicators of such enterprises. This study proposes to solve this problem by defining an algorithm using the data envelopment analysis (DEA) method. In particular, the algorithm was applied in heating systems using the example of a combined heat and power plant, where the DEA method had not previously been used. Experiments were carried out to improve the efficiency of the combined heat and power plant. Efficiency indicators were calculated, changing inputs and outputs of the model according to the study case to achieve the maximum efficiency of the system. The Charnes; Cooper and Rhodes; and the Banker, Charnes, and Cooper models were tested with good results. The presented methodology and experimental results enabled the DEA method to be applied for the first time to improve the efficiency of fuel and energy companies.



Citation: Pokushko, M.; Stupina, A.; Medina-Bulo, I.; Ezhemanskaya, S.; Kuzmich, R.; Pokushko, R. Algorithm for Application of a Basic Model for the Data Envelopment Analysis Method in Technical Systems. *Algorithms* **2023**, *16*, 460. https:// doi.org/10.3390/a16100460

Academic Editor: Frank Werner

Received: 9 August 2023 Revised: 12 September 2023 Accepted: 22 September 2023 Published: 27 September 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). **Keywords:** algorithm; data envelopment analysis; CCR model; BCC model; technical system; combined heat and power plant

1. Introduction

The significance of developing theoretical and methodological foundations for assessing the effectiveness of complex systems emerges from the need to modernize the management system of modern enterprises. Such changes are necessary due to the requirements of a rapidly changing external environment. This forms a set of new approaches to understanding the quality of an organization's activities, and consequently its viability. This becomes possible if the main criterion is met: making decisions relevant to certain conditions in which the reproduction process of the objects in the system is carried out [1–5]. Therefore, this article will discuss in detail how, depending on the goal of the decision maker, the conditions for setting the task and the results of calculating performance indicators change.

Many scientists [6–10] have addressed the problem of evaluating effectiveness. In [11], we presented a comparative analysis of all the methods of evaluating effectiveness. Among the methods analyzed was the data envelopment analysis (DEA) method, which is one of the most effective methods for evaluating the effectiveness of complex systems [12–15].

Scientists from different countries have proposed various modifications of the basic models in order to improve the DEA method [16–19]. Research in this direction is still ongoing. New publications with practical applications of the method in various fields are constantly appearing [20–24]. New DEA models are also emerging to solve the problems of evaluating the effectiveness of complex systems [25–27]. Most of the theoretical

and methodological features have been described in various scientific articles and textbooks [28–31]. In addition, scientists from different countries and scientific groups have presented various classifications of the methodology and its description. One of the method founders, William W. Cooper, created a website to highlight the development of the theoretical and practical foundations of the DEA method, as well as various innovations in the practical application of the method [32]. Furthermore, some scientists conducting research in this direction are seeking to popularize the method in scientific circles, having published textbooks, monographs, articles, and operating websites with basic information about the DEA method. There are several main theoretical sources of scientists from different countries on the models of the method and their application [33–35]. In addition, a conference dedicated to scientific research on the DEA method has been held [36]. All these sources contain various scientific works devoted to this method and the peculiarities of its application. Understanding the variety of the uses of the features, models, and modifications of basic models is somewhat complicated, with much time being required to study all the submitted works, each of which describes new scientific developments and additions to the method.

In modern scientific works, the DEA method is primarily considered when solving a variety of practical technical and economic problems [36–40]. On the basis of the practical applications, various methodologies are identified with modified models of the DEA method being used. The latest developments in this direction are described in various scientific sources [41–43]. Recently, modifications of some models have also been presented to solve certain practical problems in various industries [44–46]. Despite the interest of certain scientific groups, the popularization of the method, publications, and new developments in this field, the DEA method is still in the formation stage. Therefore, this article, to some extent, fills a gap in the scientific literature on the analysis of the effectiveness and productivity of complex systems based on the DEA method. In addition, it describes a new algorithm for applying the method in the fuel and energy complex, where the DEA method has not previously been used.

New scientific publications from different countries [47–51] have been devoted to increasing efficiency in the fuel and energy complex, focusing mainly on the introduction of new technologies and modernization of equipment for combined heat and power plants (CHPPs). The use of improved types of fuel for CHPPs is also proposed [52,53]. The efficiency of heat supply system facilities is assessed based on the performance specifications provided in quality standards and rules. Using the DEA method to improve efficiency in this area has not been presented. Therefore, we find it relevant and interesting to develop an algorithm for using the DEA method in fuel and energy complex enterprises, applying it to assess the efficiency of combined heat and power plants. This algorithm is based on DEA models.

This research seeks to solve the problem of increasing the efficiency of enterprises in the fuel and energy complex. Because such enterprises are complex systems, it is difficult to optimize their work, taking into account all the technical indicators of such enterprises. The study proposes to solve this problem using the DEA method. We modelled the cost and result indicators to achieve system efficiency. We are convinced that the algorithm and experimental results presented will allow us to apply the DEA method for the first time to improve the efficiency of fuel and energy companies.

In the continuation of the study, the methodology, the data obtained, and the conclusions from the experiments presented will form the basis of the decisions support system algorithm for the combined heat and power plant.

This paper is organized as follows: Section 1 introduces the work. Section 2 describes the basic concept of the DEA method and the basic models of the method. Section 3 then provides an overview of the main publications on the use of the DEA method, presenting a comparative analysis of the basic models and features of the use of the DEA method. The proposed methodology for applying the DEA method to fuel and energy complex enterprises is detailed in Section 4, while Section 5 presents the results of the experiments

3 of 31

on the use of the methodology to improve the efficiency of the CHPP. Finally, Section 6 presents our main conclusions.

2. Basic Concepts

The significance of developing theoretical and methodological foundations for assessing the effectiveness of complex systems emerges from the need to modernize the management system of modern enterprises. Such changes are necessary due to the requirements of a rapidly changing external environment. The management concept, which orients an organization of any type towards the effectiveness of its activities, requires the constant improvement of the existing methods and tools to enhance the efficiency of modern enterprises.

Farrell [54] was the first to propose measuring the efficiency of a complex system based on one input and one output indicator. In 1957, the author applied this model to the elements of the United States agricultural system, failing, however, to combine all the possible input and output indicators to calculate the effectiveness of the elements studied. This methodology was developed by Charnes, Cooper, and Rhodes in 1978 [55]. They suggested using mathematical programming to solve this kind of problem. The DEA method was then formulated. Over the years, the method has been widely applied in various fields to calculate both the technical and economic efficiency of the functioning of enterprises as complex systems [56–59].

At this time, the basic models of the DEA method were formulated [51]. These models have become widespread in research around the world with practical applications in various fields. Data envelopment analysis is a mathematical programming method for determining relatively useful actions of decision-making units and technical efficiency [53]. DEA helps enable the use of multiple inputs and outputs in a linear software model that calculates efficiency [53].

The system for which comparative efficiency is evaluated is called the decision-making unit (DMU) [58]. A DMU uses certain resources at the input and converts them into products of a specific type at the output. For example, it can be proposed that the DMUs under study are a CHPP. Here, it is worth clarifying the type of enterprise under consideration. This article does not consider thermal power plants designed to produce electricity for industrial purposes for external consumers but concerns thermal power plants designed for the production of heat for industrial purposes, that is, for district heating.

The combined heat and power plants considered in this article have as their main production function, like boiler houses, the production of heat (hereinafter in the calculations of the supply of thermal energy to heat networks), which is supplied to the grid for consumers. Electricity generation is an auxiliary function of such combined heat and power plants. The electricity generated is produced in relatively small quantities and is only used for the specific needs of the combined heat and power plants. Therefore, the main objective of the decision maker in the control of the CHPP is to regulate the heat supplied to the network for consumers at different times of the year. Increasing the electricity production is often impractical for such CHPPs, as it requires the construction of additional infrastructure for the consumers of the generated electricity. Therefore, we do not consider the electricity generation of these CHPPs in this study, as this function of the CHPPs under study is not the main one. Electricity is consumed only by the CHPPs themselves for their own needs.

Heat generation is the main function of such plants and satisfies the heating needs of consumers in urban agglomerates with difficult climatic conditions in cold periods. Therefore, we consider heat production as the main output of CHPPs, requiring adjustments in warm and cold seasons. As input, it uses the available thermal power of the equipment and the consumption of conventional fuel for the released fuel cell. And, as a result of the work, it has as output: the production of heat (hereinafter in the calculation of supply of thermal energy to heat networks).

In 1957, Farrell considered measuring efficiency in the presence of multiple inputs and outputs by assigning weights to inputs and outputs [54].

In this analysis, the fundamental point is efficiency, which is generally defined as the quotient of dividing the sum of all output parameters by the sum of all input factors [54]. In general, the formula is as follows [56]:

$$Te = \frac{\sum weighted \ output \ parameters}{\sum weighted \ input \ parameters}$$
(1)

where Te is technical efficiency.

Thus, efficiency according to the DEA method is generally calculated as the ratio of the sum of the weighted results of the enterprise's activities (DMU) to the sum of the weighted funds used.

For all DMUs of the studied sample of any Pareto-efficiency system, it is impossible to increase one of the outputs (decrease one of the inputs) so as not to decrease (increase) the other as a result [55].

Accordingly, when considering different tasks in improving the efficiency of the system's enterprises, different models of the DEA method are used. There are a considerable number of models. Therefore, when choosing a model of the DEA method in classical theory, the focus is typically on 3 main aspects that must be considered when constructing the DEA model [56]:

- 1. What input and output indicators of the model will be used and in what quantity, to solve the goal.
- 2. What determines the choice of a constant or variable scale effect: constant return to scale or variable return to scale.
- What determines the choice of the model orientation: output-oriented model or input-oriented model.

In practice, there may be more aspects. We will consider all the features of the model selection in more detail in the following sections of this article.

2.1. The CCR Model

The founders of the model are Charnes, Cooper, and Rhodes in 1978 [55], with the model, thus, being called the CCR model, as an abbreviation of the authors' surnames. It was one of the first to be developed, being considered a basic model in the scientific literature. Based on this model, the DEA method has been supplemented with many new models and has been further developed.

The CCR model was based on the Farrell method of measuring the efficiency of the units studied using production functions [34]. The equation with a nonlinear Farrell program displays a simple production ratio with a single input and output parameter [12–14]. In this model, multiple input and output parameters of each enterprise are combined into various scalar input and output parameters. This model is based on a constant return to scale (CRS).

It can be described by a formula that represents a solution to the maximization problem:

$$Ef = \frac{\sum_{j=1}^{s} U_j Y_{j_0}}{\sum_{i=1}^{r} V_i X_{i_0}} \to max!$$
(2)

Under the following restrictions:

$$Ef = \frac{\sum_{r=1}^{s} U_r Y_{r_m}}{\sum_{i=1}^{m} V_i X_{i_m}} \le 1$$

 $m = 1, 2, ..., n, U_r \ge 0, j = 1, 2, ..., s, V_i \ge 0, i = 1, 2, ..., r.$ where

Ef is the efficiency value of the enterprise.

n is the number of units compared.

r is the number of input factors.

s is the number of output parameters.

 x_{i_0} is an observed indicator of the i-th input factor of the enterprise.

 Y_{i_0} is an observed indicator of the j-th output parameter of the enterprise.

 X_{i_m} is the observed indicator of the i-th input factor of the m-th enterprise with i = 1,..., r and m = 1,..., n.

 Y_{i_m} is the observed indicator of the j-th output parameter of the m-th enterprise with i = 1,..., r and m = 1,..., n.

 U_r is the weighting of output factor i with I = 1,..., r.

 V_i is the weighting of input parameter j with j = 1,..., s.

When solving this maximization problem, the problem of having a quotient in the division of two linear aggregated values arises. These maximization problems are also called linear quotient programming. There are also many possibilities for transforming the linear programming of the quotient into a linear programming problem. For this purpose, Charnes, Cooper, and Rhodes, in 1978, formed the so-called inverse program of the following function [55]:

$$Ef = \frac{\sum_{i=1}^{r} V_i X_{i_0}}{\sum_{j=1}^{s} U_j Y_{j_0}} \to min!$$
(3)

Subject to the following conditions:

$$Ef = \frac{\sum_{i=1}^{m} V_i X_{i_{\mathrm{m}}}}{\sum_{r=1}^{s} U_r Y_{r_m}} \le 1$$

where m = 1, 2,..., n, $U_r \ge 0$, j = 1, 2,..., s, $V_i \ge 0$, i = 1, 2,..., r.

2.2. The BCC Model

A significant disadvantage of the CCR model discussed above is the premise of linear homogeneity [56]. Banker, Charnes, and Cooper in 1984 [13] developed a model that eliminates this drawback. It was named the BCC model, after the authors' surnames. These models differ from CCR models by adopting variable returns to scale (VRS). Variations in the BCC model make it possible to identify an increasing or decreasing scale effect for each object [60].

For the above-described linear production models, this means that if the production object (X_0, Y_0) is valid under the given economic conditions, then the object (tX_0, tY_0) where we have the number t > 0 will also be valid.

Since the concept of scale effect is important in mathematical and practical economics, we focus on it in a little more detail. It is believed that for a specific object (a point in the socio-economic space) there are the following [61]:

- A constant scale effect, if the limiting coefficient of the ratio of the relative increment of the output vector to the relative increment of the cost vector is equal to one;
- An increasing scale effect, if the limiting coefficient of the ratio of the relative increment of the output vector to the relative increment of the cost vector is greater than one;
- A decreasing scale effect, if the limiting coefficient of the ratio of the relative increment
 of the output vector to the relative increment of the cost vector is less than one.

A mathematical representation of the variable scale effect can be produced by adding a new variable U_0 to the objective function of the original model.

$$E_0 = \frac{\sum_{j=1}^{s} U_j Y_{j_0} + U_0}{\sum_{i=1}^{r} V_i X_{i_0}} \to max!$$
(4)

Subject to the following conditions:

 $\frac{\sum_{j=1}^{s} U_{j}Y_{jm} + U_{0}}{\sum_{i=1}^{r} V_{i}X_{im}} \leq 1, U_{j}, V_{i} \geq 0.$

$$U_0$$
 is the scale effect, $U_0 > 0$, $U_0 < 0$, $U_0 = 0$.

At the same time

 $U_0 > 0$ is the increasing return to scale;

 $U_0 < 0$ is the decreasing return to scale;

 $U_0 = 0$ is the constant return to scale.

The equation presented (4) is the BCC-Output Model.

ł

Now, let us imagine a multiplier primal BCC input model with variable slacks [62].

$$nax \sum_{j=1}^{s} \mu_j y_{j0} + U_0 \tag{5}$$

Subject to the following conditions:

 $\sum_{i=1}^{r} T_i X_{i0} = 1$

 $\sum_{j=1}^{s} \mu_{j} y_{j0} - \sum_{i=1}^{r} T_{i} X_{i0} + \sum_{m=1}^{n} U_{0} \le 0$

for all studied objects m = 1,..., n; $\mu_j t_j \ge 0$; U_0 is free.

The fractional program is converted to an LP problem using the Charnes and Cooper transformation and the Multiplier BCC Output model is as below [60].

$$\frac{1}{E_0} = \min \sum_{i=1}^m V_i X_{i_0} + V_0 \tag{6}$$

Subject to the following conditions:

 $\sum_{r=1}^{s} U_r Y_{r_0} = 1$

 $\sum_{i=1}^{m} V_i X_{ji} + V_0 - \sum_{r=1}^{s} U_r Y_{r_i} \ge 0$

 $j = 1, 2..., n, U_r V_i \ge \varepsilon, r = 1, 2..., s, i = 1, 2..., m, V_0$ is free (unrestricted in sign).

2.3. The Combined Heat and Power Plant

The research is designed to evaluate the efficiency of enterprises in the fuel and energy complex. Namely, the study is carried out for combined heat and power plants (CHPPs) of the heating system. Therefore, a description of the CHPPs will be presented below.

A combined heat and power plant is an enterprise whose products are electricity, as well as heat released in the form of steam and hot water, and the "raw material" is organic fuel (coal, gas, fuel oil, etc.) [49]. The equipment of the power plant is used for the economic conversion of chemical energy into electrical energy. Steam is supplied directly for industrial and household needs or is partially used for preheating water in special boilers (heaters), from which water is sent through the heating network to consumers of thermal energy. The steam condensate given to the thermal consumer is returned to the CHPP by a reverse condensate pump.

The connection of the CHPP to other stations of the power system is carried out at high voltage through step-up transformers.

Below is the composition of the combined heat and power plant:

- Fuel economy and fuel preparation system;
- Boiler plant: a combination of the boiler itself and auxiliary equipment;
- Turbine plant: steam turbine and its auxiliary equipment;
- Installation of water treatment and condensate treatment;
- Technical water supply system;
- Ash removal system (for CHP plants operating on solid fuel);
- Electrotechnical equipment and electrical equipment control system.

3. Related Work

This article considers in detail two main basic models: the CCR model and the BCC model. They are partially linear models. Scientists from different countries have developed a number of basic DEA models that have been applied in certain areas. For example, the additive DEA model (ADD) was proposed by Charnes, Cooper, Golany, Seiford, and Sturz [16] in 1985. This model does not have an input or output orientation. The ADD function of the model determines the efficiency identified from the reserve variables [59].

The Slack Based Model (SBM) is a continuation of the additive model and was introduced by Tone in 2001 [34]. It differs from the ADD model in that it is already a relative measure, not an absolute one. The Multiplicative Model was described by Charnes, Cooper, Seiford, and Sturz [16] for Cobb and Douglas production functions or functions that are partially linear-logarithmic. Thus, the efficiency frontier has a piecewise logarithmic frontier, and not a piecewise linear one, as in the case of CCR, BCC, and other models. Additionally, virtual inputs and outputs are formed multiplicatively, not additively.

3.1. Comparison of Characteristics and Criteria When Choosing the Main DEA Models: CCR and BCC

As a summary of the basic models of CCR and BCC, we present the following comparative scheme, which includes the main characteristics of these models of DEA. Figure 1 below demonstrates the comparative overview of the basic DEA CCR and BCC models.



Figure 1. Comparative overview of the basic DEA CCR and BCC models.

Thus, the basic DEA models can be classified according to the following criteria [20]:

- Type of production function;
- Selected orientation (input- or output-oriented models, as well as models without orientation);
- Acceptance of a constant (CRS) or variable return to scale (VRS).

As described above, CCR and BCC models are models with orientation to either the input or the output. The constant scale effect is applied to the model CCR. In BCC models, variable economies of scale are considered. The CCR and BCC models are partially linear models.

The BCC model is useful for variable returns to scale and measures only the net technical efficiency of each DMU [60].

The CCR model studied above considers the aggregate (technical and scale) efficiency, whereas the BCC model decomposes the total aggregate efficiency of a unit into its purely technical and scale efficiency.

To evaluate production functions with the help of DEA, it is especially important to choose a suitable DEA model for each specific case. Table 1 shows a comparison of the CCR and BCC models.

Model		Range	Type of	Da	Data	
	Scale Effect	Efficiency	Efficiency	x	У	
CCR-Output	CRS	[0.1]	TE	S-p	U-s	
CCR-Input	CRS	[0.1]	TE	S-p	U-s	
BCC-Output	VRS	[0.1]	TE	U-s	S-p	
BCC-Input	VRS	[0.1]	TE	S-p	U-s	

Table 1. Comparison of the characteristics of the described CCR and BCC models.

CRS—Constant Return to Scale, VRS—Variable Return to Scale, TE—Technical Efficiency, U-s—unrestricted in sign, S-p—Semi-p (= semi-positive)—only positive values are allowed.

Thus, Table 1 shows that the CCR and BCC models are used to measure technical efficiency. For all basic models, efficiency is measured in the range from 0 to 1.

In scientific articles, CCR and BCC models have a wide range of practical applications. They have been used to evaluate the efficiency of enterprises in economics, banking, agriculture, medicine, industry, trade, transport, and many others. However, the use of DEA models in the fuel and energy complex has not been investigated.

We next take a closer look at the main characteristics that need to be considered when choosing a DEA model.

3.2. Comparison of Characteristics for the DEA Method

When choosing a DEA model, the first decision is whether to use a constant scale effect (CRS) or variable scale effect (VRS). Upon closer examination, it can be said that the output parameters change proportionally to the input parameters during the CRS. When the VRS changes the output parameters, the output parameters may change disproportionately [60].

Färe, Grosskopf, and Levell [63] divide technical efficiency into two related multiplier components. The division is based on the dependence of efficiency on the magnitude of the scale. Considering enterprises for returns on scale, we can speak of the dependence of productivity on an increase in the resources used for production. If the increase in productivity is proportional to the increase in the number of resources used, then the enterprise has a CRS.

An enterprise achieves the highest possible level of productivity if it is 100% efficient, with both a constant and a variable return on scale.

If an enterprise is 100% efficient with VRS, but fails to achieve full efficiency with CRS, there then arises the efficiency depending on the magnitude of scale—scale efficiency (SE).

The formula of this regularity is as follows:

$$TE = SE \times PTE \tag{7}$$

where

TE—technical efficiency with constant return of scale;

PTE—technical efficiency with variable return of scale, or pure technical efficiency; SE—efficiency depending on the magnitude of the scale.

When calculating these performance indicators, it is possible to identify the sources of inefficiency. In the case of PTE, inefficiency may be caused by the inefficient operation of the enterprise. Then, in the case of SE, inefficiency can be caused by unfavorable conditions. In the experimental section, we will show this with examples.

Figure 2 below demonstrates the difference between the three performance indicators (TE, SE, and PTE) discussed above.



Figure 2. Efficiency depending on the magnitude of the scale [8].

In Figure 2, the input parameters are located along the *X*-axis. The *Y*-axis shows the output parameters.

The direct line is a line on which 100% efficient enterprises with a constant return of scale should be located. On the F curve, there are 100% efficient enterprises with variable returns of scale. Thus, enterprise A, lying on the curve F, is fully efficient with VRS (PTE = 1), but not efficient enough with CRS. Therefore, the efficiency, depending on the magnitude of the scale, is determined by the ratio [64]

$$SE(A) = \frac{LM}{LA} < 1 \tag{8}$$

Both enterprise B and enterprise C are 100% effective at both CRS and VRS, so they operate at the highest possible level of productivity. For an inefficient enterprise E, efficiency can be determined using the following formula:

$$SE(E) = \frac{PQ}{PE} = \frac{PE}{PR}$$
(9)

Figure 2 and the calculations show that the inefficiency of enterprise E is caused by both inefficient production and unfavorable conditions.

4. Materials and Methods of using DEA Models to Calculate the Efficiency of Fuel and Energy Complex Enterprises

In the previous paragraphs, the basic DEA models were investigated. We now consider the possibility of applying these models in practice in a specific area and build an algorithm for using DEA models to calculate the efficiency of fuel and energy complex enterprises. To do this, we will follow the next steps for evaluating the effectiveness of the DEA method:

- Studying the scope in which it is planned to evaluate the effectiveness of the studied objects (CHPP);
- 2. Setting a goal;
- 3. Selection of the investigated CHPP;
- 4. Analysis of the external environment and basic CHPP operations;
- 5. Selection of input and output parameters (evaluation factors);
- 6. Selection of the model and its orientation in accordance with the goal;
- 7. Software selection;
- 8. Interpretation of the results obtained;
- 9. Recommendations for improving efficiency.



Based on these steps, we have compiled a diagram of an algorithm for evaluating the effectiveness of the DEA method. It is shown in Figure 3.

Figure 3. Scheme of the algorithm for evaluating the effectiveness of the DEA method.

This algorithm can be applied to many complex systems. In this article, we have applied it to a complex technical system. This is justified by the fact that in the management of technical systems, decision makers often face the difficulties of regulating constituent objects while improving the efficiency of the system as a whole. The measurement of a large number of system performance indicators, as well as their regulation, often creates problems in the management of the system as a whole. We applied the algorithm in the district heating system. This is justified by the global limitation of energy and heat resources across the world. This system, in our opinion, is subject to the emergence of a number of problems in this regard, as, in accordance with global strategies, energy and heat should be saved. The district heating system, in our view, requires the application of the efficiency-improvement method, which has hitherto not been applied in this system as it will contribute to the reduction of resource consumption and increase the efficiency of enterprises in the system. The experimentation of the algorithm proposed in the article is carried out for CHPPs in the district heating system. The following reasons underlie the choice of these enterprises and the system itself:

- The global strategy of scientific and technological development of fuel and energy companies;
- The need to modernize the management system of modern fuel and energy companies in accordance with the requirements of a rapidly changing external environment;

- The aim of increasing competitiveness in enterprises;
- The transition of fuel and energy companies to advanced digital, intelligent production technologies.

In addition, one of the most important problems for the fuel and energy sector today is that of ecology and environmental pollution. For more than a decade, residents of cities where combined heat and power plants are located have been forced to bear unfavorable living conditions associated with pollution. Such emissions also cause damage to the global environment as a whole. The global strategy of conservation and reduction of environmental pollution dictates the policy of the world powers to maximize the production of heat and electricity with minimal damage to the planet.

As noted above, the main methodologies recently proposed by scientists have focused on the introduction of new technologies and modernization of equipment for combined heat and power plants. The use of improved types of fuel for CHPPs is also proposed. The efficiency of the heat supply system facilities is assessed based on the performance specifications specified in the quality standards and rules. The algorithm proposed in the article for using the DEA method is based on a different approach, which does not involve the replacement of equipment or technology. Therefore, it requires no additional high financial investment. The proposed methodology allows us to consider only external indicators (input and output indicators) to assess effectiveness. This simplifies calculations, unlike the situation of joint analysis of internal and external indicators. The proposed algorithm allows for the flexibility and high adaptability of the system in accordance with changes in environmental factors and changes in the purpose of the decision maker. Using the DEA method allows us not only to evaluate the effectiveness of the enterprises analyzed, but also to configure the inputs and outputs of non-effective enterprises to achieve efficiency.

Below, we describe this algorithm in more detail for this field of application.

Next, in the experimental part, we will consider six experiments. In these experiments, we use the CCR model and the BCC model. In the first four experiments, we will use the following input and output data:

- (1) Available thermal power of the equipment (Gcal/hour)—input (x1).
- (2) The consumption of conventional fuel for the released fuel cell released (thousand tons/year)—input (x2)
- (3) Supply of thermal energy to heat networks (thousand Gcal)—output (y1).In experiment №5, we will use the following inputs and outputs:
- (1) The available thermal power of the equipment (Gcal/hour)—input (x1).
- (2) The consumption of conventional fuel per released fuel cell (thousand tons/year) input (x2).
- (3) Supply of thermal energy to heat networks (thousand Gcal)—output (y1).
- (4) The mass of the emission (thousand tons per year)—output (y2).

In experiment №6, we used the following inputs and outputs:

- (1) Heating capacity of turbo generators (Gcal/hour)—input (x1).
- (2) Power of peak hot water boilers (Gcal/hour)—input (x2).
- (3) Power of steam boilers (Gcal/hour)—input (x3).
- (4) Consumption of conventional fuel for the released fuel (thousand tons/year)—input (x4).
- (5) Heat supply to the grid (thousand Gcal)—output (y1).

4.1. Selecting the Main Criteria for Assessing Effectiveness

For this study, the operation of the district heating system and the technological processes of the CHPP plant were studied.

The main output indicator of the CHPP is the release of thermal energy to heat networks [20]. Taking into account the environmental factor, which was described above, it is reasonable to consider also an output indicator: the mass of emissions. However, if the inclusion of the environmental aspect is not required for the decision-maker (DM) when

achieving a specific goal, then the model can be built with one input indicator: the release of thermal energy into the network.

Thus, as input indicators in this study, we use the following:

- (1) The available thermal power of the equipment (Gcal/hour).
- (2) The consumption of conventional fuel per released fuel cell (thousand tons/year). As output indicators, we use the following:
- (1) Supply of thermal energy to heat networks (thousand Gcal).
- (2) The mass of the emission (thousand tons per year).

There may be situations when there is a need to reduce part of the output indicators, that is, it is necessary to reduce the mass of emissions from boilers. To do this, for calculations, we do not use the actual values of the output indicators but their deviations from the threshold value set at a level obviously exceeding the values of the corresponding indicators for all objects studied.

Furthermore, in order to see how to regulate the available thermal power for the main types of equipment, it is more expedient to divide it into power for each type of main equipment. Therefore, this indicator is divided into the following three indicators:

- Heating capacity of turbo generators (Gcal/hour);
- Capacity of peak hot water boilers (Gcal/hour);
- Power of steam boilers (Gcal/hour).

In this study, we conduct experiments on various variations of input and output indicators. It should be noted that DEA models have limitations on the number of observations to obtain reliable calculated data. This is reflected in many scientific studies. Accordingly, in the literature, it is accepted that the number of observations is calculated using the following formula [64]:

$$N > max\{M x S, 3(M+S)\}$$

$$\tag{10}$$

where

N is the number of CHPs, *M* is the number of input indicators, and *S* is the number of output indicators.

Therefore, to conduct experiments, it is advisable to choose the required number of observations, no fewer than what is obtained for this sample when calculating according to the formula presented above [64].

In a real calculation situation, if the number of observations is lower than that determined by the formula, this problem can be solved in two ways. For example, this can be achieved by increasing the number of CHPPs. If it is impossible to increase the number of CHPPs investigated, then the scientific literature describes the possibility of using observations on the same CHPPs in a different time period, taking them as new CHPPs [64].

Depending on the goals of the decision maker, the orientation of the DEA model may be different. The model can be input-oriented if the goal is to reduce the input parameters, and if the goal is to increase output indicators, then we use an output-oriented model.

In our study, we conduct experiments on two types of orientation: input and output. Thus, we can see the difference and draw conclusions about the feasibility of each option.

We perform experiments using the main models: CCR and BCC. We conduct experiments with both a constant scale effect and a variable one to see if there are any matches on the effectiveness of the CHPP from the analyzed sample at VRS and CRS. We, thus, see what causes the inefficiency of the enterprises studied: inefficient work and/or unfavorable conditions.

We have chosen the Charnes, Cooper, and Rhodes model and the Banker, Charnes, and Cooper model because these models are the basic models of the DEA method. Since these models are the basis for the operation and use of the DEA method. In order to use additional DEA models, we first need to analyse the effectiveness of the basic models. We have chosen these models as models of widespread use. Often, the models discussed in this article are sufficient to analyse the efficiency of technical systems. We have chosen these models so that readers can understand the essence of the DEA method. And, in the future, they can also try to use additional DEA models and easily analyse using the basic models.

4.2. Calculating, Interpreting Results, and Making Recommendations to Improve Efficiency

We now describe the software for conducting experiments. There is currently a large number of software products based on DEA models. The most common include DEEP, Master, DEAD Frontier, PIM-DEA Soft, DEAS, OpenDEA, DEAOS, EMS, DEA Solver Online, Frontier Analyst, DEA-SolverPro, and KonSi—Data coverage analysis for comparative analysis. A comparative study of such software was conducted by Iliyasu et al. [53].

Our review of all these software capabilities led to the selection of DEEP as the most suitable software for research on CCR and BCC models, being that used in this study [65].

Nonetheless, the existing software products are unfortunately limited only to assessing the current level of efficiency, without offering means to explain this level, issuing recommendations on ways to achieve the required level of efficiency (in particular, by redistributing resources between subsystems).

When using the calculated results, the question of the interpretation of the results obtained arises, as well as the development of recommendations for improving efficiency in this field. In addition, it is difficult to use these software tools without a detailed study of the DEA method and the software support itself. Existing software products do not use "large" database management systems—Oracle, PostgreSQL, MySQL, etc. The data is stored in Excel tables, in a Microsoft Access (mdb) database, and in text files. Accordingly, engineers and other decision makers often cannot correctly use the models of the DEA method, using the presented software each time to solve a specific goal. Accordingly, the actual use of such software in practice for solving problems in complex technical systems is limited. For specialists in the field of computer technology, solving problems in complex technical systems is difficult without specialist knowledge and experience in the field under study. It is also difficult for them to correctly form input and output parameters, as well as the model itself, its orientation and scale, as the technical systems are complex. In order to be able to evaluate their effectiveness, a detailed study of such systems functioning is required. In addition, recommendations should also contain information on how, and according to what, input and output indicators can be changed, as the program outputs only the optimal combination of input and output values to maximize efficiency in the sample of studied enterprises. It is, thus, difficult to interpret the results obtained. Accordingly, how to achieve the recommended indicators should be described in the expert's recommendations.

This problem may be solved by a programmed DSS at the enterprises studied. Accordingly, to automate the formation of result interpretation when programming a DSS, a large number of preliminary experiments are required.

Therefore, the next section is devoted to conducting such experiments and analyzing the data obtained.

5. Results

We now present the experimental part of the study. In this section, we test the methodology given in the previous paragraph for using basic DEA models to calculate the efficiency of fuel and energy complex enterprises. We investigate changes in the sample indicators and changes in models, their orientation, inputs, and outputs when achieving different goals of the DM. The objects of the study will be combined heat and power plants. A total of 27 CHPPs will be represented in the sample [66]. This is a sufficient number of CHPPs to conduct experiments and ensure the correctness of calculations using the DEA method. In the presentation of research results, in order to simplify the visualization of the data obtained, only a part of the sample is presented. We next present the results for six CHPPs in order to describe and analyze the main results and patterns obtained in the experiments.

To build a basic model, for initial experiments, we take the simplest combination of inputs and outputs: two inputs and one output to demonstrate the possibility of testing this method in the field studied. Let us imagine the input and output parameters. Inputs:

- (1) Available thermal power of the equipment (Gcal/hour)—input (x1).
- (2) The consumption of conventional fuel for the fuel cell released (thousand tons/year) input (x2).

Outputs:

 Supply of thermal energy to heat networks (thousand Gcal)—output (y1). The initial data are presented in Table 2.

№ СНРР	O (y1)	I (x1)	I (x2)
1	3635	1554	642
2	4519	1530	790
3	1077	752	401
4	3626	1554	641
5	4380	1405	767
6	1001	752	386

Table 2. Initial data on two input indicators and one output indicator of the studied CHPP sample.

O-output; I-input.

If it is important for the decision maker to increase the supply of thermal energy to heat networks, it is advisable to use the output-oriented DEA model. If it is important for the decision maker to reduce the consumption of conventional fuel for the released fuel cell and reduce the available thermal power of the equipment, then it is advisable to use the input-oriented DEA model.

We next conduct experiments on different conditions and goals for the decision maker. We calculate the efficiency using the DEA method in DEEP software [67] (http://www.uq. edu.au/economics/cepa/deap.php (accessed on 22 May 2022)).

5.1. Experiment 1—The Output-Oriented CCR Model with Two Input and One Output Parameters

Next, we build a basic output-oriented CCR model. The goal for this experiment is to increase the supply of thermal energy to heat networks.

The results obtained are presented in Table 3.

Table 3. Efficiency calculation based on the output-oriented CCR model with two input and one output parameters.

№ СНРР	Eff	O (y1)	I (x1)	I (x2)
1	0.873	4161.898	1554	642
2	0.978	4620.025	1530	790
3	0.466	2312.812	752	401
4	0.872	4158.590	1554	641
5	1.000	4380.000	1405	767
6	0.442	2263.190	752	386

Eff—effectiveness.

It can be seen that only CHPP 5 has an efficiency indicator of 1 and is, thus, effective. It does not require setting the output indicator. Therefore, in the table, the output indicator y1 remains unchanged for CHPP 5. For all other CHPPs, the y1 indicator changes. Table 3 shows the y1 values required to achieve the effectiveness of this CHPP. For example, the efficiency indicator calculated by the DEA method for CHP1 is 0.873 with the maximum possible being 1. The input indicator output (y1) is initially equal to 3635.

In order to increase the efficiency of this CHPP1 to 1, and so make the enterprise fall on the efficiency frontier built on this sample, it is necessary to increase output (y1) to 4161,898. The efficiency coefficient calculated by the DEA method will be equal to 1, and, accordingly, the enterprise will become efficient.

Input indicators input (x1) and input (x2) remain unchanged when the model is oriented to the output of the CCR model; only the output indicator changes.

5.2. Experiment 2—The Input-Oriented CCR Model with Two Input and One Output Parameters

We now build a basic input-oriented CCR model. The goal for this experiment is to reduce the consumption of conventional fuel for the released fuel cell and reduce the available thermal power of the equipment. The results obtained are presented in Table 4.

Table 4. Efficiency calculation based on the input-oriented CCR model with two input and one output parameters.

№ CHPP	Eff	O (y1)	I (x1)	I (x2)
1	0.873	3635	1357.263	560.723
2	0.978	4519	1496.544	772.725
3	0.466	1077	350.182	186.732
4	0.872	3626	1354.980	558.907
5	1.000	4380	1405.000	767.000
6	0.442	1001	332.607	170.726

From this table it can be seen that the performance indicators have not changed relative to experiment 1. In this case, only the orientation of the model has changed. Now the DEA model minimizes the input metrics without changing the output metric.

As can be seen from Table 4, in experiment 1, only CHPP 5 has an efficiency index of 1 and is, thus, effective. It does not require setting the output indicator. Therefore, in the table, the indicators of inputs (x1) and (x2) remain unchanged for CHPP 5. For all the other CHPPs, the values of inputs (x1) and (x2) change. Table 4 shows the values (x1) and (x2) required to achieve the effectiveness of this CHPP. For example, the efficiency indicator calculated using the DEA method for CHPP 1 is 0.873 with the maximum possible being 1. The input indicator (x1) is initially equal to 1554 and the input indicator (x2) is initially equal to 642. In order to increase the efficiency of this CHPP 1 to 1, and so make the enterprise fall on the efficiency frontier built on this sample, it is necessary to reduce input (x1) to 1357.263; input (x2) should be reduced to 560.723. In this case, the efficiency coefficient calculated using the DEA method will be equal to 1 and, accordingly, the company will become efficient.

The output indicator (y1) remains unchanged when oriented to the input of the CCR model. Only the indicators of the inputs change.

5.3. Experiment 3—The Input-Oriented BCC Model with Two Input and One Output Parameters

We now build a basic input-oriented BCC model. The goal for this experiment is to reduce the consumption of conventional fuel for the fuel cell released and reduce the available thermal power of the equipment. However, as described above, the BCC model is a VRS model. We make calculations using the same initial data from Table 3 as for the CCR model. Accordingly, we will see how the performance indicators will change and what the scale effect will be for each CHPP.

The results obtained are presented in Table 5.

Table 5. Efficiency calculation based on the input-oriented BCC model with two input and one output parameters.

№ СНРР	Eff	O (y1)	I (x1)	I (x2)
1	0.893	3635	1388.020	573.429
2	1.000	4519	1530.000	790.000
3	0.761	1311	572.000	292.000
4	0.891	3626	1385.218	571.380
5	1.000	4380	1405.000	767.000
6	0.761	1311	572.000	292.000

From this table, it can be seen that, according to this model, 2 enterprises are already effective: CHPPs 2 and 5. They have an efficiency score of 1, and so are effective. They do not require setting up input indicators. Therefore, in Table 5, the values of inputs (x1) and (x2) remain unchanged for CHPPs 2 and 5. For all the other CHPPs, the values of inputs (x1) and (x2) change. Accordingly, Table 5 shows the values of inputs (x1) and (x2) required to achieve the efficiency of the remaining CHPPs. For example, the efficiency indicator calculated using the DEA model BCC method for CHP 1 is 0.893 with the maximum possible being 1. The input indicator input (x1) is initially equal to 1554 and the input indicator input (x2) is initially equal to 642. In order to increase the efficiency of this CHPP1 to 1, and so make the enterprise fall on the efficiency frontier built on this sample, it is necessary to reduce input (x1) to 1388.020. The input (x2) indicator is reduced to 573.429. In this case, the efficiency coefficient calculated using the DEA model to 1 and, accordingly, the company will become efficient.

The output indicator (y1) according to the BCC model for CHPP 1 has not changed. In this model, unlike the CCR model, the output indicator can also change in the presence of slack movement for the output indicator. In the section describing the models, this phenomenon was described in sufficient detail. Slack movement can be used for both output indicators and input indicators. In our sample, such CHPPs are present, and there is slack movement for both input and output indicators.

Table 6 presents the slack movement for the indicators of the sample studied.

№ CHPP	O (y1)	I (x1)	I (x2)
1	-	-	-
2	-	-	-
3	234	-	-13.016
4	-	-	-
5	-	-	-
6	310	-	-1.606

Table 6. Slack movement for the indicators of the input-oriented BCC model with two input and one output parameters.

As we can see from Table 6, slack movement is present for CHPPs 3 and 6. The CHPP data tends towards the CHPP 27 indicators, with CHPPs 3 and 6 tending to the nearest effective frontier point, which for them is CHPP 27. Accordingly, slack movement appears.

As described above, the BCC model, unlike the CCR model, helps to calculate efficiency considering variable return to scale, while determining technical efficiency and scale

efficiency. Therefore, to analyze all performance indicators for this sample, it is advisable to present a comparison of all the performance indicators for the sample studied. Below we present a table comparing performance indicators for the output-oriented CCR and BCC models with two input and one output parameter.

As we can see from Table 7, for this sample, only CHPP 5 achieves maximum efficiency since CRATE, CRATE and SE are equal to 1. CHPP 2 achieves an efficiency factor of 1 only with variable returns on scale. With a constant return of scale, the efficiency coefficient is 0.978; SE for CHP 2 is also 0.978. In this case, we can say that the sources of inefficiency for this enterprise are caused by unfavorable conditions. These sources are not caused by the inefficient operation of the enterprise, since the technical efficiency for this enterprise is equal to 1.

Table 7. Comparative table of performance indicators for input-oriented CCR and BCC models with two input and one output parameters.

СНРР	CRSTE	VRSTE	SE	Scale Type
1	0.873	0.893	0.978	drs
2	0.978	1.000	0.978	drs
3	0.466	0.761	0.612	irs
4	0.872	0.891	0.978	drs
5	1.000	1.000	1.000	-
6	0.442	0.761	0.581	irs

CRSTE—constant return of scale technical efficiency, VRSTE—variable return of scale technical efficiency, SE—scale efficiencies, DRS—decreasing return of scale, IRS—increasing return to scale, CRSTE—technical efficiency from CRS DEA, VRSTE—technical efficiency from VRS DEA, Scale—scale efficiency = CRSTE/VRSTE.

Moreover, in the table, we can see a comparison of the performance indicators CRSTE, VRSTE, and SE for the remaining CHP, and we can also determine the type of return on scale: decreasing return to scale or increasing return to scale.

For example, CHPP 1 has the following performance indicators: CRSTE = 0.873, VRSTE = 0.893, and SE = 0.978. Analyzing these performance indicators, we can state the following: The sources of inefficiency for this enterprise are caused more by the inefficient operation of the enterprise than by unfavorable conditions, since VRSTE = 0.893, which is significantly less than 1. The indicator SE = 0.978, which is almost equal to 1. The most inefficient enterprises from the analyzed sample are CHPPs 3 and 6. Moreover, as can be seen from Table 7, their inefficiency is caused by both the inefficient operation of the enterprise and unfavorable conditions.

Analyzing the type of return of scale for this sample, we can say that CHPPs 3 and 6 are at the point of return to scale. The same CHPPs as described above have slack movements. CHPPs 1, 2, and 4 are at the point of decreasing return to scale.

5.4. Experiment 4—The Output-Oriented BCC Model with Two Input and One Output Parameters

We now build a basic output-oriented BCC model. The goal for this experiment will be to increase the supply of thermal energy to heat networks.

We calculate the efficiency using the output-oriented BCC model with two input and one output parameters.

This table shows that, according to this model, two enterprises are also effective: CHPPs 2 and 5, having an efficiency score of 1. They do not require setting an output score. Therefore, in the table, the output indicator (y1) remains unchanged for CHPPs 2 and 5. For all other CHPPs, the (y1) indicator changes. Table 8 shows the (y1) values required to achieve the efficiency of the CHPP data. For example, the efficiency indicator calculated using the DEA method for CHPP 1 is 0.925 with the maximum possible being 1. The output indicator (y1) is initially equal to 3635.

№ СНРР	Eff	O (y1)	I (x1)	I (x2)
1	0.925	3931.163	1472.188	642.000
2	1.000	4519.000	1530.000	790.000
3	0.546	1974.169	752.000	394.641
4	0.923	3927.191	1554.000	641.000
5	1.000	4380.000	1405.000	767.000
6	0.514	1949.111	752.000	386.000

Table 8. Efficiency calculation based on the output-oriented BCC model with two input and one output parameters.

In order to increase the efficiency of this CHPP 1 to 1, and so make the enterprise fall on the efficiency frontier built on this sample, it is necessary to increase output (y1) to 3931.163. In this case, the efficiency coefficient calculated using the DEA method will be equal to 1, and, accordingly, the enterprise will become efficient.

The input (x1) indicator for the BCC model also changed from 1554 to 1472.188 as, according to the input indicator (x1), there was a slack movement equal to -81.813. The input indicator (x2) remains unchanged.

Table 9 presents the slack movement for the indicators of the studied sample.

Table 9. Slack movement for the indicators of the output-oriented BCC model with two input and one output parameters.

№ CHPP	O (y1)	I (x1)	I (x2)
1	_	-81.813	-
2	-	-	-
3	-	-	-6.359
4	-	-82.203	-
5	-	-	-

Table 9 shows slack movement is present for CHPPs 1, 3, and 4. The data for CHPPs 1 and 4 tend towards the indicators of CHPP 26. CHPP 3 tends towards the indicators of CHPP 27. Thus, CHPPs 1 and 4 tend to the nearest effective frontier point, which is CHPP 26 for them. CHPP 3 tends to the nearest effective frontier point, which for it is CHPP 26. Accordingly, this slack movement appears.

Below we present a table comparing performance indicators for the output-oriented CCR and BCC models with two input and one output parameters.

As the table shows, for this sample CHPP 5 is an effective enterprise according to the presented performance indicators: CRSTE, VRSTE, and SE. It achieves maximum efficiency in its work. CHPP 2 achieves an efficiency factor of 1 only with variable returns on scale. With a constant return on scale, the efficiency coefficient is 0.978. SE for CHPP 2 is also 0.978. In this case, we can say that the sources of inefficiency for this enterprise are caused by unfavorable conditions, and are not caused by the inefficient operation of the enterprise, since the technical efficiency for this enterprise is equal to 1.

All performance indicators for the remaining CHPPs are below 1 and are also presented in Table 10. For example, CHPP 1 has the following performance indicators: CRSTE = 0.873and VRSTE = 0.925, SE = 0.945. Analyzing these performance indicators, we can say the following: The sources of inefficiency for this enterprise are caused more by the inefficient operation of the enterprise than by unfavorable conditions. The most inefficient enterprises from the sample analyzed, as in the previous experiment, are CHPPs 3 and 6. Moreover, as can be seen from Table 7, their inefficiency is caused by both the inefficient operation of the enterprise and unfavorable conditions.

CHPP	CRSTE	VRSTE	SE	Scale Type
1	0.873	0.925	0.945	DRS
2	0.978	1.000	0.978	DRS
3	0.466	0.546	0.854	IRS
4	0.872	0.923	0.944	DRS
5	1.000	1.000	1.000	-
6	0.442	0.514	0.861	IRS

Table 10. Comparative table of performance indicators for output-oriented CCR and BCC models with two input and one output parameters.

In the previous experiment, it was observed that CHPPs 1, 2, and 4 were at the point of decreasing scale effect (decreasing return to scale). CHPPs 3 and 6 were at the point of increasing scale effect (increasing return to scale).

5.5. Experiment 5—The Output-Oriented BCC Model with Two Input and Two Output Parameters

We conducted experiments on the basic CCR models with an input and output orientation with two inputs and one output. Conclusions were drawn from these experiments about the objects of the studied sample. There was also a comparison of performance indicators, and we calculated indicators of inputs and outputs when using different DEA models and their orientations.

Depending on the purpose of the DM, inputs, and outputs, models and their orientation may vary. It was previously mentioned that the environmental component is important when evaluating the efficiency of the CHP. Therefore, we believe that it is advisable to add another output, depending on the purpose of the decision maker, that is, whether it is important in these conditions to regulate the mass of emissions into the atmosphere. Then, it is also possible to use a model with two inputs and two outputs, adding as an output an indicator of accounting for emissions into the environment. Namely, as input indicators, we also use the following:

- (1) The available thermal power of the equipment (Gcal/hour)—input (x1).
- (2) The consumption of conventional fuel per released fuel cell (thousand tons/year) input (x2).

As output indicators we use the following.

- (1) Supply of thermal energy to heat networks (thousand Gcal)—output (y1).
- (2) The mass of the emission (thousand tons per year)—output (y2).

We now build a basic output-oriented BCC model. The goal for this experiment will be to increase the supply of thermal energy to heat networks and reduce the mass of the emission.

We calculate the efficiency using the output-oriented BCC model with two input and two output parameters.

We do not give the full results of the CCR model experiments separately but present only a comparative table of performance indicators for the CCR and BCC models. In addition, as we have seen from previous experiments, our sample contains CHPPs that have decreasing and increasing scale effects. Accordingly, it is more expedient to measure efficiency using the BCC model for this sample.

In Table 11, the O (y2) column shows the values of the outputs (y2) emission mass (thousand tons per year) for each CHP. In accordance with the selected orientation of the model, the output indicators will increase. However, according to the goal set, the emission mass index needs to be reduced. As described above, in such cases, the deviation indicator from the maximum value of this indicator is used for the entire sample. In our case, the maximum value is 17.4, which, rounded up to an integer, will be 18. Next, we present

rounded indicators of deviation from the maximum indicator for each CHPP. The data is presented in Table 12.

№ CHPP	O (y1)	O (y2)	I (x1)	I (x2)
1	3635	14.9	1554	642
2	4519	13.5	1530	790
3	1077	7.30	752	401
4	3626	15.2	1554	641
5	4380	13.9	1405	767
6	1001	7.90	752	386

Table 11. Initial data on two input indicators and two output indicators of the studied CHPP sample.

Table 12. Initial data (deviation coefficients) for the output (y2) of the studied CHPP sample.

CHPP 1 (y2)	CHPP 2 (y2)	CHPP 3 (y2)	CHPP 4 (y2)	CHPP 5 (y2)	CHPP 6 (y2)
3	5	11	3	4	10

The indicators presented in Table 12 will be used to calculate the effectiveness of the BCC model.

We present the results of efficiency calculations based on the BCC model in Table 13.

Table 13. Efficiency calculation based on the output-oriented BCC model with two input and two output parameters.

№ СНРР	VRSTE	y1	y2	x1	x2
1	0.925	3931.163	4.075	1472.188	642
2	1.000	4519.000	5.000	1530.000	790
3	1.000	1180.000	11.000	752.000	355
4	0.923	3927.191	4.069	1471.797	641
5	1.000	4380.000	4.000	1405.000	767
6	0.909	1180.000	11.000	752.000	355

According to this model, three enterprises are effective: CHPPs 2, 3, and 5. They have an efficiency indicator of 1. They do not require setting output indicators. Therefore, in the table the outputs indicators (y1) and (y2) remain unchanged. For all other CHPPs, the indicator (y1) and (y2) changes.

In Table 13, the values of outputs (y2) should be explained. The output (y2) values for CHPPs 2, 3, and 5 have not changed. Accordingly, the outputs (y2) change for CHPPs 1, 4, and 6. The changes can be explained. For CHPP 1 it is required to reduce (y2) by 1.075; for CHPP 4 it is required to reduce (y2) by 1.069; and for CHPP 6 it is required to reduce (y2) by 1. At the same time, the outputs (y1) for these CHPPs will also change to increase the efficiency indicator to 1. We now explain the changes in outputs (y1). For CHP 1 it is required to increase y2 by 296.163; for CHPP 4 it is required to increase (y2) by 301.191; and for CHPP 6 it is required to increase (y2) by 179.

The input parameters (x1) and (x2) according to the BCC model have also changed for some CHPPs due to slack movement. For CHPPs 2 and 5, slack movement is not observed in any indicator. For the rest of the CHPPs, such changes are present. For CHPP 1, as a result of adjusting indicators to achieve an efficiency coefficient of 1, slack movement occurs with output (y2) and input (x1). Output (y2) decreases by 0.831, and input (x1) decreases by 81,812. For CHPP 3, as a result of adjusting indicators to achieve an efficiency coefficient of 1, slack movement occurs with output (y1) and input (x2). Output (y1) increases by 103 and input (x2) decreases by 46. For CHPP 4, as a result of adjusting indicators to achieve an efficiency coefficient of 1, slack movement occurs with output (y2) and input (x1). Output (y2) decreases by 0.82, and input (x2) decreases by 82.203. For CHPP 6, as a result of adjusting indicators to achieve an efficiency coefficient of 1, slack movement occurs with output (y1) and input (x2). Output (y1) increases by 78.9 and input (x2) decreases by 31.

As described above, the BCC model helps to calculate efficiency considering VRS. The CCR model calculates efficiency using CRS. Below, we present a comparative table of performance indicators for the output-oriented CCR and BCC models with two input and one output parameters.

As we can see from the table, for this sample, CHPPs 2, 3, and 5 are effective enterprises according to the performance indicators of CRSTE, VRSTE, and SE. That is, they achieve maximum efficiency in their work. CHPP 6 achieves an efficiency factor of 1 only at SE. This means that for this enterprise, unfavorable conditions have no effect on its efficiency. With a constant return on scale and a variable return on scale, the efficiency coefficients are the same and equal to 0.909. Thus, the inefficiency of this enterprise is wholly caused by the inefficient operation of the enterprise. All performance indicators for the remaining CHPPs are below 1 and are also presented in Table 14. Nonetheless, as we can see from the table, the performance indicators for all the enterprises in the sample are considerably high.

Table 14. Comparative table of performance indicators for output-oriented CCR and BCC models with two input and two output parameters.

CHPP	CRSTE	VRSTE	SE	Scale Type
1	0.892	0.925	0.965	DRS
2	1.000	1.000	1.000	-
3	1.000	1.000	1.000	-
4	0.891	0.923	0.965	DRS
5	1.000	1.000	1.000	-
6	0.909	0.909	1.000	-

In this experiment, we observe changes in the display of the scale effect. From Table 14, it can be seen that only CHPPs 1 and 4 are at the point of decreasing scale effect (decreasing return to scale).

In this experiment, it can be seen that the efficiency coefficients are significantly higher for the same sample as in previous experiments. There are also significantly more efficient enterprises. In comparison, with the previous experiment, where CHPPs 3 and 6 had a rather low efficiency of CRSTE and VRSTE (below 0.6), CHPP 3 has now become an efficient enterprise, and CHP P6 has CRSTE and VRSTE efficiency indicators of more than 0.9. The introduction of another output, which the model focuses on, yields a significant change in the indicators for the sample. Depending on the purpose of the decisionmaker, the input and output indicators are adjusted and the efficiency coefficients are changed according to the DEA model.

5.6. Experiment 6—The Input -Oriented BCC Model with Four Input Indicators and One Output Indicator

We conducted experiments on the basic models of BCC and CCR with an input and output orientation, using two inputs and one output and two inputs and two outputs, respectively. From these experiments, conclusions were drawn about the objects of the sample studied and changes in performance indicators.

Changing inputs and outputs can occur when the target of the decision maker changes. It is expedient to reflect certain inputs and outputs, when solving a certain task of increasing efficiency. In the study of the CHPP of the heating system, it is possible to replace the input of the available thermal power of the equipment with three inputs: the heating capacity of turbo generators (Gcal/hour); the power of peak hot water boilers (Gcal/hour); and the power of steam boilers (Gcal/hour). This will give us the opportunity to adjust the efficiency of work for each type of equipment. Depending on the purpose of the decision maker, you can also use a model with four inputs and one output. Namely, as input indicators we use the following:

- (1) Heating capacity of turbo generators (Gcal/hour)—input (x1).
- (2) Power of peak hot water boilers (Gcal/hour)—input (x2).
- (3) Power of steam boilers (Gcal/hour)—input (x3).
- (4) Consumption of conventional fuel for the released fuel (thousand tons/year)—input (x4). As an output indicator, we use the following:
- (1) Heat supply to the grid (thousand Gcal)—output (y1).

This combination of inputs and outputs will help adjust the efficiency of the equipment and adjust fuel consumption and heat release when achieving maximum efficiency of the CHPP.

The initial data are presented in Table 15.

and one output indicator.

Table 15. Initial data on four input indicators and one output indicator of the CHPP sample.

№ СНРР	O (y1)	I (x1)	I (x2)	I (x3)	I (x4)
1	3635	850	210	230	642
2	4519	909	226	270	790
3	1077	270	440	42	401
4	3626	850	210	230	641
5	4380	909	226	270	767
6	1001	270	440	42	386

We now build a basic input-oriented BCC model. The goal for this experiment will be to reduce the power of equipment and fuel consumption while maintaining the indicator of heat supply to the grid. We will calculate the efficiency using the input-oriented BCC model with four input indicators and one output indicator of the studied CHPP sample. We present the results obtained in Table 16.

 Table 16. Efficiency calculation based on the input-oriented BCC model with four input indicators

№ СНРР	O (y1)	I (x1)	I (x2)	I (x3)	I (x4)
1	3635	850.000	210.000	230.0	595.681
2	4519	909.000	226.000	270.0	790.000
3	1077	190.000	340.000	42.0	283.902
4	3626	850.000	210.000	230.0	593.122
5	4380	899.769	223.705	263.9	759.211
6	1001	190.000	340.000	42.0	281.271

We also present a table of performance indicators for the input-oriented CCR and BCC models with four input and one output parameters, providing a visual opportunity to see changes in input indicators depending on performance indicators.

As can be seen from Table 17, according to the BCC VRSTE model, five enterprises are effective: CHPPs 1, 2, 3, 4, and 6. They have an efficiency score of 1. However, for four of them (CHPPs 1, 3, 4, and 6) there is slack movement. Only for CHPP 2 do the input indicators not change, with slack movement also not being observed for any indicator. For the rest of the CHPPs, such changes are present.

СНРР	CRSTE	VRSTE	SE	Scale Type
1	0.957	1.00	0.957	irs
2	1.000	1.00	1.000	-
3	0.627	1.00	0.627	irs
4	0.956	1.00	0.956	irs
5	0.989	0.99	0.999	irs
6	0.582	1.00	0.582	irs

Table 17. Comparative table of performance indicators for input-oriented CCR and BCC models with four input indicators and one output parameter.

For CHP 1, as a result of adjusting indicators to achieve an efficiency coefficient of 1, slack movement occurs with input (x1) and is 46.319. For CHPP 3, as a result of slack movement, input (x1) decreases by 80, input (x2) decreases by 100, and input (x4) decreases by 117.098. For CHPP 4, as a result of adjusting indicators to achieve an efficiency coefficient of 1, slack movement occurs with input (x1) and is 46.319. For CHPP 5, as a result of slack movement, input (x3) decreases by 3.341. Finally, for CHPP 6, as a result of adjusting indicators to achieve an efficiency factor of 1, slack movements occur with input (x1), which decreases by 80; input (x2) decreases by 100 and input (x4) decreases by 104,729.

As can be seen from Table 16, the input (x1-x4) indicators for CHPPs 1, 2, 3, 4, and 6 have not changed. As described above, only slack movements occurred there. For CHPP 5, the input indicators (x1-x4) change. Input (x1) decreases by 9.231, input (x2) decreases by 2.295, input (x3) decreases by 2.742, and input (x4) decreases by 7.789.

As we can see from Table 17, for this sample, only CHPP 2 is an effective enterprise. According to the presented performance indicators of CRSTE, VRSTE, and SE, CHPP 2 achieves maximum efficiency in operation. All performance indicators for the remaining CHPPs for CRSTE and SE are below 1 and are also presented in the table. Table 17 shows CHPPs 1, 3, 4, 5, and 6 are at the point of increasing return to scale.

As can be seen from Table 17, the performance indicators for CHPPs 1, 3, 4, and 6 are equal to 1, but the performance indicators of CRSTE and SE are below one. This suggests that these enterprises are working quite efficiently. Hence, the inefficiency of these enterprises is wholly caused by unfavorable conditions. These have a particularly strong effect on CHPPs 3 and 6, since their SE is 0.627 and 0.582, respectively.

In this experiment, it can be seen that the efficiency coefficients for VRSTE are significantly higher for the same sample as in previous experiments, being 1 for five CHPPs. However, only one CHP 2 achieves maximum efficiency in operation. If in previous experiments CHPP 5 was effective and achieved maximum efficiency in operation, then, in this experiment, it is the only CHPP that has a VRSTE efficiency coefficient below 1. As can be seen from Table 17, in this experiment five CHPPs are at the point of increasing scale effect (increasing return to scale), while, in experiments 3 and 4 only CHPPs 3 and 5 were at the point of increasing return to scale.

6. Discussion

In this article, we investigated the theoretical aspects of evaluating the effectiveness of complex systems, as well as conducting an analysis of literary sources on the subject of the study. The concept of efficiency and productivity of complex systems based on the DEA method was analyzed. The DEA method itself is described in sufficient detail. The basic DEA models are described and their comparative analysis is presented. The methodology of applying the DEA method in a specific field was presented. We also demonstrated the application of this methodology in the fuel and energy complex, a context in which the DEA method had not previously been used. The application of the methodology for assessing complex technical systems was presented in the district heating system using the example of the CHPP. A total of 27 CHPPs were represented in the study sample,

for which experiments were conducted to improve the efficiency of the CHPP with the setting of different DM goals. As a result, when the goal was changed, different DEA models were built. Calculations of efficiency indicators were carried out when changing inputs and outputs, according to the orientation of the model for the same studied sample. The indicators of inputs and outputs were calculated to achieve maximum efficiency—an efficiency indicator equal to 1. The CCR and BCC models were tested, with an input orientation and an output orientation. The models were built according to variations of inputs and one output, two inputs and two outputs, and four inputs and one output, respectively. The following performance indicators were calculated in the experiments: technical efficiency from CRS DEA, technical efficiency from VRS DEA, and scale efficiencies. It was also determined at which point of the scale effect (decreasing return to scale/increasing return to scale) each of the enterprises in the studied sample is located.

The article presents 6 experiments. The first four experiments were carried out on the sample analyzed with two inputs (the available thermal power of the equipment and the consumption of conventional fuel for the released fuel cell) and one output (the release of thermal energy to heat networks). In these experiments, efficiency calculations were carried out using the CCR and BCC models. We compared the results of efficiency calculations using these models. We compared the inputs and outputs of inefficient DMUs when they achieved efficiency. We analyzed the reasons for inefficiency in different experiments. In the experiments of the paper we detailed the results using these models. The limitations of using these models are described in the paper.

When setting the goal of reducing the consumption of conventional fuel for the released fuel cell and reducing the available thermal power of the equipment, the following results were obtained. The input-oriented DEA model was set. Under experiments 1 and 3, CHPP 5 is an effective enterprise according to the presented performance indicators: CRSTE, VRSTE, and SE. That is, it achieves maximum efficiency in its work. CHPP 2 achieves an efficiency factor of 1 only with VRS. With a CRS, the efficiency coefficient is 0.978. SE for CHPP 2 is also 0.978. In this case, we can say that the sources of inefficiency for this enterprise are caused by unfavorable conditions. However, their influence is low, since the efficiency coefficient is quite high and is equal to 0.978. This enterprise works reasonably efficiently, since the technical efficiency for this enterprise is also equal to 1. The most inefficient enterprises from the sample are CHPPs 3 and 6. Moreover, their inefficiency is caused by both the inefficient operation of the enterprise and unfavorable conditions. For all inefficient enterprises, the required input indicators were calculated to achieve an efficiency coefficient equal to 1. Analyzing the type of return on scale according to these experiments, we can say that CHPPs 3 and 6 are at the point of increasing return to scale, while CHPPs 1, 2, and 4 are at the point of decreasing return to scale. Slacks movements are observed for CHPP 3 and 6 for the BCC model. When calculating using the CCR model, slacks movements are not observed.

In experiments 2 and 4, the goal was to increase the supply of thermal energy to heat networks. To this end, efficiency calculations were carried out and the following results were obtained. An output-oriented DEA model was set. Under experiments 2 and 4, CHPP 5 is an effective enterprise according to the presented performance indicators: CRSTE, VRSTE, and SE. The same results were obtained for experiments 1 and 3. Identical efficiency coefficients were also obtained for CHPP 2. For the rest of the enterprises in the sample, the CRSTE performance indicators are also identical, while the performance indicators of VRSTE and SE differ. The most inefficient enterprises from the sample, as in the previous experiment, are CHPPs 3 and 6. Their inefficiency is also caused by both inefficient operation of the enterprise and unfavorable conditions. For all inefficient enterprises, the required output indicators were calculated to achieve an efficiency coefficient equal to 1.

In the experiment, it is observed that CHPPs 1, 2, and 4 are at the point of decreasing return to scale. CHPP 3 and 6 are at the point of increasing return to scale. Slack movements

were already observed for CHPPs 1, 3, and 4 for the BCC model. For experiments 1 and 3, only CHPPs 3 and 6 presented slack movements, while no slack movements are found in calculations using the CCR model.

In experiment 5, another output indicator was added. This was done in order to see the impact of the environmental factor on the efficiency of the CHPP, adjusting the performance of the CHPP plant while reducing emissions into the atmosphere. In this experiment, the goal was to increase the supply of thermal energy to heat networks and reduce the mass of harmful substances released into the atmosphere. In accordance with the goal set, an output-oriented model with two inputs and two outputs was used. As input indicators, the following were used: the available thermal power of the equipment, the consumption of conventional fuel for the released fuel cell. As output indicators, the following were used: the release of thermal energy to heat networks and the emission mass. The efficiency calculations using the DEA method for CCR and BCC models yielded the following data. In this experiment, with the same sample, three enterprises are already effective enterprises according to the performance indicators of CRSTE, VRSTE, and SE, namely CHPPs 2, 3, and 5. They achieve maximum efficiency in their work. CHPP 6 achieves an efficiency factor of 1 only at SE. This means that for this enterprise, unfavorable conditions have no effect on its efficiency. With a constant return on scale and a variable return on scale, the efficiency coefficients are the same and equal to 0.909. Thus, the inefficiency of this enterprise is wholly caused by the inefficient operation of the enterprise. All performance indicators for the remaining CHPPs are below 1. For these, the required output indicators were calculated to achieve an efficiency coefficient equal to 1.

In this experiment, we observe changes in the display of the scale effect. Now, only CHPPs 1 and 4 are at the point of decreasing scale effect (decreasing return to scale). In addition, for CHPPs 2 and 5 slack movement is not observed on any indicator. For the rest of the CHPPs such changes are present.

In this experiment, it can be seen that the efficiency coefficients are significantly higher for the same sample as in previous experiments. Additionally, there are significantly more efficient enterprises. In comparison with the previous experiment, where CHPPs 3 and 6 had a rather low efficiency of CRSTE and VRSTE (below 0.6), CHPP 3 is now an efficient enterprise. CHPP 6 has CRSTE and VRSTE performance indicators of more than 0.9. Accordingly, the introduction of another output, which the model focuses on, yielded a significant change in the performance indicators for the sample. In this case, with the introduction of the emission mass output, there was a significant increase in efficiency indicators. This may indicate that enterprises in the field of environmental factors are working quite efficiently. However, in the absence of this output, the same enterprises have lower performance indicators. All these features should be considered when compiling the algorithm of the DSS, adding possible options for changing the goal when solving the problem of increasing efficiency by the DEA method.

In experiment 6, the input of the available thermal power of the equipment was replaced by three inputs: the heating power of turbo generators, the power of peak hot water boilers, and the power of steam boilers. This was done in order to be able to adjust the efficiency of work for each type of equipment used at the CHPP. In this experiment, the goal was to reduce the power of the equipment and the consumption of conventional fuel for the released fuel cell. Accordingly, the DEA model with four inputs and one output, oriented to the input, was used. As input indicators, the following were used: the heating capacity of turbo generators, the power of peak hot water boilers, the power of steam boilers, and the consumption of conventional fuel for the released fuel cell. As an output indicator, we used the heat energy released to the grid. This combination of inputs and outputs helped adjust the efficiency of each type of equipment, adjusting fuel consumption when achieving maximum efficiency of the CHPP. As a result of efficiency calculations using the DEA method for CCR and BCC models, the following data were obtained. In this experiment, with the same sample, only CHPP 2 is an effective enterprise according to the performance indicators of CRSTE, VRSTE, and SE. That is, CHPP 2 achieves maximum

efficiency in operation. In addition, in this experiment, 5 enterprises are effective according to VRSTE: CHPPs 1, 2, 3, 4, and 6, having an efficiency score of 1. However, for four of them (CHPPs 1, 3, 4, and 6), there is slack movement. Only for CHPP 2 do the input indicators not change, while slack movement is also not observed for any indicator. For the rest of the CHPPs, such changes are present.

The efficiency indicators for CHPP 1, 3, 4, and 6 are equal to 1. However, the efficiency indicators of CRSTE and SE are below 1. This suggests that these enterprises are working quite efficiently. The inefficiency of these enterprises is wholly caused by unfavorable conditions. They have a particularly strong effect on CHPP 3 and CHPP 6, since their SE is 0.627 and 0.582, respectively.

In this experiment, it can be seen that the efficiency coefficients for VRSTE are significantly higher for the same sample as in previous experiments. For five CHPPs, these are 1, but only one CHPP, number 2, achieves maximum efficiency in operation. If in previous experiments CHPP 5 was effective and achieved maximum efficiency in operation, then in this experiment it is the only CHPP that has a VRSTE efficiency coefficient below 1. In this experiment, five CHPPs are at the point of increasing scale effect (increasing return to scale). Meanwhile, in experiments 3 and 4 only CHPPs 3 and 5 were at the point of increasing scale effect (increasing return to scale).

Now, let us compare the data from experiment 3 and this experiment. In experiment 3 the goal was the same, but only in this experiment, the input indicator of the available thermal power of the equipment was divided into three indicators: the heating capacity of turbo generators, the power of peak hot water boilers, and the power of steam boilers. This was done to make it possible to regulate the power of each type of equipment at the CHPP. When comparing the performance indicators according to these experiments, it is clear there is a significant difference, both in terms of the scale effect and in the efficiency of the enterprises in the same sample. For example, the most efficient enterprise in experiment 3 was CHPP 5, while, in this experiment, it is CHPP 2. The VRSTE efficiency is equal to 1 for CHPPs 1, 3, 4, 5, and 6 in this experiment. In experiment 3, the VRSTE efficiency was equal to 1 only for CHPPs 2 and 5.

The inefficiency of enterprises in this experiment is mainly caused by unfavorable conditions. The inefficiency of enterprises in experiment 3 is mainly caused by both unfavorable conditions and inefficient operation of enterprises. In addition, in this experiment, all enterprises in the sample have an increasing return to scale. In contrast, in experiment 3, only CHPPs 3 and 6 have an increasing return to scale. The remaining CHPPs are at the point of decreasing return to scale.

The limitations of the algorithm were not revealed. When using this algorithm, you should accurately understand the purpose of the decision maker and at the expense of which tools of the DEA model the goal can be achieved. In the experimental part, we examined in detail the various options for goals and how they can be solved using the DEA method. It is also necessary to know how the investigated DMUs function in order to correctly select the inputs and outputs.

The algorithm can be applied to other technical systems.

Readers do not need additional parameters to conduct such experiments. Data sources are needed—data on the inputs and outputs of the DMU. No more data are required. Readers can repeat the experiments described in the article without problems, since they are analyzed in sufficient detail.

Using the DEA method and conducting experiments makes it possible to program a decision support system. This will make it possible to apply this algorithm to enterprises in the form of a decision support system and increase the adaptability and manageability of enterprises of the fuel and energy complex.

7. Conclusions and Future Work

In this article, we have described in sufficient detail the algorithm of using the DEA method for district heating enterprises. Therefore, readers can easily use the results of these experiments and repeat such calculations independently.

Our experiments have demonstrated how much the performance indicators of the CHPP can change. In accordance with these indicators, the settings of input and output indicators for the sample are also changed to achieve efficiency. For example, for one experiment, the same CHPP can achieve maximum efficiency in operation, and for another it can be ineffective. When the goal to be achieved changes the conditions for the DEA task change. The results of calculations of the effectiveness of DEA models for the same sample may change dramatically.

It can be concluded that for the possibility of application, as well as for the correctness of the application of the DEA method, an understanding of the application of this method to various models in a particular field is required. As well as understanding how and when the goal changes, it is necessary to understand that the indicators of inputs and outputs, the model itself, and the orientation of the DEA model change. The district heating system is a rather complex system with a large number of technical indicators. Therefore, achieving the efficiency of such systems is a complex and time-consuming process. The introduction of new methods for evaluating the effectiveness of such systems will solve a number of problems. In the future, it will lead to automation of management processes, due to the possibility of creating a DSS to increase efficiency based on the DEA method. It will also lead to an increase in the adaptability of the system to external factors due to the possibility of changing the settings of the DEA method when changing the goals of the DM.

The data obtained from the experiments made it possible to observe changes in performance indicators when changing the goal of DM in the same sample. It was clearly demonstrated that the use of various DEA models and their orientations leads to changes in the settings of input and output indicators. Thus, only with a full understanding of the patterns and performance indicators of the studied field is the correct use of DEA models possible. The objectives of the DM may change in the course of the functioning of the enterprises due to rapidly changing external factors in the operating environment. Therefore, in these conditions, it is important to change the settings for building DEA models for specific enterprises, as well as their input and output parameters. The methodology of using DEA models for evaluating efficiency presented in this article makes it possible to apply the DEA method to improve the efficiency of fuel and energy companies, a context in which this method and methodology have not previously been used.

In the article, we have shown how, when the goal of the decision maker changes, the performance indicator can change. This is a very important conclusion for understanding the change in efficiency when the operating conditions of the system change. Now, it has become clear how we could adjust the effectiveness when changing the goal of the decision maker. The results obtained in the experimental part will be the basis for programming an automated decision-making system in the future.

There are 2 aspects on which we can continue working:

1. To develop an automated decision support system for fuel and energy complex enterprises based on the algorithm and experimental results.

2. When using the DEA method, there are several usage restrictions. Further work can be continued on the development of a new efficiency method that can minimize the limitations of its use.

The algorithm presented, the data obtained and the conclusions from the presented experiments will help in using the DEA method to increase the efficiency of fuel and energy companies. In the continuation of this study, they will form the basis of the DSS algorithm for combined heat and power plants. In further research, an automated DSS will be developed based on the results obtained. This DSS will be based on an algorithm for applying different DEA models depending on the performance of the CHP. It will also be possible to integrate the DSS based on the DEA method into the management system of

modern CHPPs. This DSS will help automatically calculate and adjust the desired change in the performance of the system elements from changes in the indicators of inputs and outputs over time and depending on changes in external conditions. This will automate the process of increasing the efficiency of fuel and energy companies based on the DEA method.

Author Contributions: M.P., A.S. and I.M.-B. were engaged in the formulation of research goals and objectives, conducted a simulation of the algorithm for using DEA models to calculate the efficiency of fuel and energy complex enterprises, organized experiments, and prepared a publication. S.E. organized the resource provision of information and reference materials for research and was engaged in the preparation and editing of the publication. R.K. and R.P. were responsible for statistics, were engaged in data storage and processing, and carried out data analysis, processing, and visualization. They engaged in the design of the publication. All authors have read and agreed to the published version of the manuscript.

Funding: This research was founded by MCIN/AEI/10.13039/501100011033/ and ERDF A way to do Europe grant number AwESOMe Project PID2021-122215NB-C33 and in part by the Ministry of Science and Higher Education of the Russian Federation grant number 075-15-2022-1121. The APC was funded by MCIN/AEI/10.13039/501100011033/ and ERDF A way to do Europe grant number AwESOMe Project PID2021-122215NB-C33.

Data Availability Statement: The data used in the paper are publicly available at the following link: Administration of the City of Krasnoyarsk. The Project "Heat Supply Scheme of the City of Krasnoyarsk Until": [Electronic Resource]. 2015. Available online: http://www.admkrsk.ru/citytoday/municipal/energy/teplosn/Documents/%D0%9F%D1%80%D0%BE%D0%B5%D0%BA%D1%82%20%D1%81%D1%85%D0%B5%D0%B5%D0%B5%D0%B5%D0%B6%D1%81%D0%BD%D0%B6%D0%B5%D0%B5%D0%B6%D0%B6%D0%B5%D0%B6%D0%B5%D0%B7%D0%B6%D0%B5%D0%B7%D0%D7%D0%D7%D0%B7%D0%B7%D0%D

Acknowledgments: This work was supported by the Ministry of Science and Higher Education of the Russian Federation (Grant № 075-15-2022-1121) and in part by the Spanish Ministry of Science and Innovation and the European Regional Development (ERDF) Funds (Grant PID2021-122215NB-C33).

Conflicts of Interest: The authors declare no conflict of interest.

References

- Geida, A.S.; Ismailova, Z.F.; Klitnyu, I.V.; Lysenko, I.V. Relations of effectiveness of system functioning, system potential and adjacent operational properties. *Spiiras Proc.* 2015, 4, 2–18. [CrossRef]
- Fisun, K.; Naumenko, M.; Bortnyk, N. Price of the effiency of functioning of organizational systems using expert methods. *Balt. J. Econ. Stud.* 2018, 4, 325–332. [CrossRef]
- Agafonova, N.P.; Medvedeva, E.A.; Serdyukov, A.A.; Kempf, A.A.; Litvinova, O.G. Opportunities and limitations of application of management accounting tools in practical activity of economic entities. In *Advances in Science, Technology and Innovation*; Springer Nature: Berlin/Heidelberg, Germany, 2022; pp. 329–334. [CrossRef]
- Latysheva, V.M.; Postnikova, L.V.; Postnikova, D.D.; Kovaleva, N.N.; Ivanova, O.P. Analysis of the contract system functioning effectiveness. In Proceedings of the International Scientific and Practical Conference "Russia 2020—A New Reality: Economy and Society" (ISPCR 2020), Veliky Novgorod, Russia, 9–10 December 2020; p. 164. [CrossRef]
- 5. Houck, M.M. Effectiveness. Encycl. Forensic Sci. 2022, 3, 206–209. [CrossRef]
- 6. Jablonský, J. Efficiency analysis in multi-period systems: An application to performance evaluation in Czech higher education. *Cent. Eur. J. Oper. Res.* **2015**, *24*, 283–296. [CrossRef]
- 7. Uri, N.D. Measuring the impact of incentive regulation on technical efficiency in telecommunications in the United States. *Appl. Math. Model.* **2004**, *28*, 255–271. [CrossRef]
- 8. Taymaz, E.; Saatci, G. Technical change and efficiency in Turkish Manufacturing Industries. J. Product. Anal. 1997, 8, 461–475. [CrossRef]
- Solow, R.M. Investment and technical progress. Mathematical methods in the social sciences. In *First Stanford Symposium* 1959; Stanford University Press: Stanford, CA, USA, 1960; pp. 89–104.
- Morrison, C.J.; Nehring, R.; Banker, D.; Breneman, V. Productivity growth, technological progress, and technical efficiency in the heartland and Southern Cotton States: 1996–1999. In Proceedings of the AAEA Annual Meetings, Chicago, IL, USA, 5–8 August 2001.
- Pokushko, M.; Stupina, A.; Medina-Bulo, I.; Dresvianskii, E.; Mashinets, E.; Shmeleva, Z. Applying the Data Envelopment Analysis method for evaluating the efficiency of the complex system operations in fuel and energy companies. *J. Phys. Conf. Ser.* 2020, 1515, 052039. [CrossRef]

- 12. Charnes, A.; Haag, S.; Jaska, P.; Semple, J. Sensitivity of efficiency classifications in the additive model of data envelopment analysis. *Int. J. Syst. Sci.* **1992**, *23*, 789–798. [CrossRef]
- Sexton, T.R.; Silkman's, R.H.; Hogan's, A.J. Data Envelopment Analysis: Critique and Extensions. New Dir. Program Eval. 1986, 19, 73–105. [CrossRef]
- 14. Selford, L.M. Data Envelopment Analysis: The evolution of the state of the art (1978–1995). J. Product. Anal. 1996, 7, 99–137.
- 15. Banker, R.D.; Charnes, A.; Cooper, W.W. Some models for estimating technical and scale inefficiencies in Data Envelopment Analysis. *Manag. Sci.* **1984**, *30*, 1078–1092. [CrossRef]
- 16. Charnes, A.; Cooper, W.W.; Seiford, L.M.; Sturz, J. A multiplicative model for efficiency analysis. *Socio-Econ. Plan. Sci.* **1982**, *16*, 223–224. [CrossRef]
- Seiford, L.M.; Zhu, J. Infeasibility of super-efficiency Data Envelopment Analysis models. *Inf. Syst. Oper. Res.* 1999, 37, 174–187. [CrossRef]
- Sharp, J.A.; Meng, W.; Liu, W. A modified slacks-based measure model for data envelopment analysis with 'natural' negative outputs and inputs. J. Oper. Res. Soc. 2007, 58, 1672–1677. [CrossRef]
- Chernyshova, G.Y.; Kovalev, R.N. Application of Data Envelopment Analysis to assess the effectiveness of web-resources. *Manag. Sci.* 2017, 41, 1172–1184. [CrossRef]
- Wang, X.; Lu, Y.; Chen, C.; Yi, X.; Cui, H. Total-factor energy efficiency of ten major global energy-consuming countries. J. Environ. Sci. China 2024, 137, 41–52. [CrossRef]
- Gidion, G.K. Investigation of water utility efficiency features: Production variables and the DEA model choices. *Water Supply* 2023, 23, 1700–1710. [CrossRef]
- 22. Michali, M.; Emrouznejad, A.; Dehnokhalaji, A.; Clegg, B. Subsampling bootstrap in network DEA. *Eur. J. Oper. Res.* 2023, 305, 766–780. [CrossRef]
- Sekitani, K.; Zhao, Y. Least-distance approach for efficiency analysis: A framework for nonlinear DEA models. *Eur. J. Oper. Res.* 2023, 306, 1296–1310. [CrossRef]
- Omrani, H.; Shamsi, M.; Emrouznejad, A. Evaluating sustainable efficiency of decision-making units considering undesirable outputs: An application to airline using integrated multi-objective DEA-TOPSIS. *Environ. Dev. Sustain.* 2023, 25, 5899–5930. [CrossRef]
- 25. Ruiyue, L. Cross-efficiency evaluation capable of dealing with negative data: A directional distance function based approach. *J. Oper. Res. Soc.* **2020**, *71*, 505–516.
- Adhikari, A.; Basu, S.; Biswas, I.; Banerjee, A.; Pratim, P.; Sengupta. A route efficiency analysis using Shannon entropy-based modified DEA method and route characteristics investigation for urban bus transport in India. *INFOR Inf. Syst. Oper. Res.* 2018, 3, 332–359. [CrossRef]
- 27. Ruiyue, L.; Chen, Z. A directional distance based super-efficiency DEA model handling negative data. J. Oper. Res. Soc. 2017, 68, 1312–1322.
- Dehnokhalaji, A.K.; Emrouznejad, A.A. A box-uncertainty in DEA. A robust performance measurement framework. *Expert Syst. Appl.* 2022, 187, 115855. [CrossRef]
- 29. Zhu, N.; Emrouznejad, A.A. A combined machine learning algorithms and DEA method for measuring and predicting the efficiency of Chinese manufacturing listed companies. *J. Manag. Sci. Eng.* **2021**, *6*, 435–448. [CrossRef]
- 30. Zhang, G.J.; Cui, J.C. A general inverse DEA model for non-radial DEA. Comput. Ind. Eng. 2020, 142, 106368. [CrossRef]
- Castelli, L.; Pesenti, R.; Ukovich, W. A classification of DEA models when the internal structure of the Decision-Making Units is considered. Ann. Oper. Res. Ann. Oper. Res. 2008, 173, 207–235. [CrossRef]
- 32. Banker, R.D.; Kaplan, R.S. William W. Cooper: Innovator, Fighter, and Scholar. Account. Horiz. 2014, 28, 193–203. [CrossRef]
- 33. Wu, J.; Chu, J.; Zhu, Q.; Li, Y.; Liang, L. Determining common weights in data envelopment analysis based on the satisfaction degree. *J. Oper. Res. Soc.* **2016**, *67*, 1446–1458. [CrossRef]
- 34. Cooper, W.W.; Seiford, L.M.; Tone, K. Data Envelopment Analysis; Kluwer Academic Publishers: Boston, MA, USA, 2000; p. 318.
- Krivonozhko, V.E.; Lychev, A.V. Analysis of the Activity of Complex Economic Systems; Publishing Department of the Faculty of VMiK Lomonosov Moscow State University: Moscow, Russia; Max Press: London, UK, 2010; p. 208.
- Call for PAPERS: Data Envelopment Analysis (DEA) Stream at IFORS2020, 21–26 June 2020, Seoul, Korea. Available online: http://deazone.com/en/ifors2020 (accessed on 29 January 2023).
- Hermoso-Orzáez, M.J.; García-Alguacil, M.; Terrados-Cepeda, J.; Brito, P. Measurement of environmental efficiency in the countries of the European Union with the enhanced data envelopment analysis method (DEA) during the period 2005–2012. *Environ. Sci. Pollut. Res.* 2020, 27, 15691–15715. [CrossRef]
- Rebolledo-Leiva, R.; Angulo-Meza, L.; Iriarte, A.; González-Araya, M.C.; Vásquez-Ibarra, L. Comparing two CF+DEA methods for assessing eco-efficiency from theoretical and practical points of view. *Sci. Total Environ.* 2019, 659, 1266–1282. [CrossRef] [PubMed]
- Hajduk, S. Efficiency evaluation of urban transport using the dea method. Equilibrium. Q. J. Econ. Econ. Policy 2018, 13, 141–157. [CrossRef]

- 40. Babacan, A.; Bulut, Ş.; Koç, N. Efficiency analysis of the banks operating in turkey with AHP based on Dea method. *Amfiteatru Econ.* **2020**, *22*, 882–903. [CrossRef]
- 41. Pascoe, S.; Cannard, T.; Dowling, N.A.; Dichmont, C.M.; Asche, F.; Little, L.R. Use of Data Envelopment Analysis (DEA) to assess management alternatives in the presence of multiple objectives. *Mar. Policy* **2023**, *148*, 105444. [CrossRef]
- 42. Kyrgiakos, L.S.; Kleftodimos, G.; Vlontzos, G.; Pardalos, P.M. A systematic literature review of data envelopment analysis implementation in agriculture under the prism of sustainability. In *Operational Research*; Springer Science and Business Media Deutschland GmbH: Berlin/Heidelberg, Germany, 2023. [CrossRef]
- 43. Ivanová, E.; Grmanová, E. Digital transformation and ICT sector performance in EU countries. *Probl. Perspect. Manag.* 2023, 21, 48–58. [CrossRef]
- 44. Kang, C.C.; Feng, C.M.; Chou, P.F.; Wey, W.M.; Khan, H.A. Mixed network DEA models with shared resources for measuring and decomposing performance of public transportation systems. *Res. Transp. Bus. Manag.* **2023**, *46*, 100828. [CrossRef]
- Ryandono, M.N.H.; Widiastuti, T.; Cahyono, E.F.; Filianti, D.; Qulub, A.S.; Al Mustofa, M.U. Efficiency of zakat institutions in Indonesia: Data envelopment analysis (DEA) vs free disposal hull (FDH) vs super-efficiency DEA. J. Islam. Account. Bus. Res. 2023. ahead-of-print. [CrossRef]
- 46. Boubaker, S.; Le, T.D.Q.; Ngo, T.; Manita, R. Predicting the performance of MSMEs: A hybrid DEA-machine learning approach. In *Annals of Operations Research*; Springer: Berlin/Heidelberg, Germany, 2023. [CrossRef]
- Tamburini, A.; Cipollina, A.; Micale, G.; Piacentino, A. CHP (combined heat and power) retrofit for a large MED-TVC (multiple effect distillation along with thermal vapour compression) desalination plant: High efficiency assessment for different design options under the current legislative EU framework. *Energy* 2016, 115, 1548–1559. [CrossRef]
- Zhu, S.; Yu, G.; Liang, K.; Dai, W.; Luo, E. A review of Stirling-engine-based combined heat and power technology. *Appl. Energy* 2021, 294, 116965. [CrossRef]
- Wu, J.; Li, R.; Xiang, Y.; Wang, S.; Wang, L.; Li, Q. Quantitative sustainability assessment and sensitivity analysis for a solar-driven combined heating and power system integrated with organic Rankine cycle and ground source heat pump. *Appl. Therm. Eng.* 2023, 220, 119722. [CrossRef]
- 50. Kasaeian, A.; Nouri, G.; Ranjbaran, P.; Wen, D. Solar collectors and photovoltaics as combined heat and power systems: A critical review. In *Energy Conversion and Management*; Elsevier Ltd.: Amsterdam, The Netherlands, 2018. [CrossRef]
- 51. Elmer, T.; Worall, M.; Wu, S.; Riffat, S.B. Fuel cell technology for domestic built environment applications: State of-the-art review. In *Renewable and Sustainable Energy Reviews*; Elsevier Ltd.: Amsterdam, The Netherlands, 2015. [CrossRef]
- 52. Vakalis, S.; Moustakas, K. Applications of the 3T method and the R1 formula as efficiency assessment tools for comparing waste-to-energy and landfilling. *Energies* **2019**, *12*, 1066. [CrossRef]
- 53. Ciuła, J.; Generowicz, A.; Gaska, K.; Gronba-Chyła, A. Efficiency Analysis of the Generation of Energy in a Biogas CHP System and its Management in a Waste Landfill—Case Study. *J. Ecol. Eng.* **2022**, *23*, 143–156. [CrossRef]
- 54. Farrell, M.J. The measurement of productive efficiency. J. R. Stat. Soc. Ser. A-G 1957, 120, 253–290. [CrossRef]
- 55. Charnes, A.; Cooper, W.W.; Rhodes, E. Measuring the efficiency of decision-making units. *Eur. J. Oper. Res.* **1978**, *2*, 429–444. [CrossRef]
- 56. Emrouznejad, A.; Yang, G. A survey and analysis of the first 40 years of scholarly literature in DEA: 1978–2016. *Socio-Econ. Plan Sci.* **2018**, *61*, 4–8. [CrossRef]
- 57. Foroughi, A.A.; Shureshjani, R.A. Solving generalized fuzzy data envelopment analysis model: A parametric approach. *Cent. Eur. J. Oper. Res.* 2017, *25*, 889–905. [CrossRef]
- 58. Roháčová, V. A DEA based approach for optimization of urban public transport system. *Cent. Eur. J. Oper. Res.* **2015**, *23*, 215–233. [CrossRef]
- 59. Takano, Y.; Ishii, N.; Muraki, M. Multi-period resource allocation for estimating project costs in competitive bidding. *Cent. Eur. J. Oper. Res.* 2017, 25, 303. [CrossRef]
- 60. Banker, R.; Charnes, A.; Cooper, W. An introduction to data envelopment analysis with some of its models and their uses. Researches in governmental and non-profit accounting. *Eur. J. Oper. Res.* **1989**, *1*, 125–163.
- 61. Cook, W.; Zhu, J. Modeling Performance Measurement Applications and Implementation Issues in DEA; Springer: New York, NY, USA, 2005; p. 408.
- Lissitsa, A.; Babiéceva, T. Data Envelope Analysis (DEA)-a modern methodology for determining production efficiency. Discussion, Institute of Agricultural Development in Central and Eastern Europe (IAMO), Halle (Saale) and Eastern Europe (IAMO), 50. 2003. Available online: http://nbn-resolving.de/urn:nbn:de:gbv:3:2-23263 (accessed on 15 February 2023).
- 63. Färe, R.; Grabowski, R.; Grosskopf, S.; Kraft, S. Efficiency of a fixed but allocable input: A non-parametric approach. *Econ. Lett.* **1997**, *56*, 187–193. [CrossRef]
- 64. Cooper, W.W. Origin and Development of Data Envelpment Analysis: Challenges and Opportunities. *Data Envel. Anal. J.* **2014**, *1*, 3–10. [CrossRef]
- 65. Iliyasu, A.; Mohamed, Z.A. Evaluating contextual factors affecting the technical efficiency of freshwater pond culture systems in Peninsular Malaysia: A two-stage DEA approach. *Aquac. Rep.* **2016**, *3*, 12–17. [CrossRef]

- 66. Administration of the City of Krasnoyarsk. The Project "Heat Supply Scheme of the City of Krasnoyarsk Until": [Electronic Resource]. 2015. Available online: http://www.admkrsk.ru/citytoday/municipal/energy/teplosn/Documents/%D0%9 F%D1%80%D0%BE%D0%B5%D0%BA%D1%82%20%D1%81%D1%85%D0%B5%D0%BC%D1%8B%20%D1%82%D0%B5%D0 %BF%D0%BB%D0%BE%D1%81%D0%BD%D0%B0%D0%B1%D0%B6%D0%B5%D0%BD%D0%B8%D1%8F.pdf (accessed on 22 May 2022).
- 67. CEPA Centre for Efficiency and Productivity Analysis. Software: [Electronic Resource]. Available online: http://www.uq.edu. au/economics/cepa/deap.php (accessed on 22 May 2022).

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.